Handout 3 Feb. 6 1 6.894 OODL Design and Implementation Spring 2001 Massachusetts Institute of Technology Department of Electrical Engineering and Computer Science

Assignment #1

Due: (via email) Thursday, 2/8/01.

There is an online version of this form under the class website: http://www.ai.mit.edu/projects/dynlangs/oodl-course/spring01/assign1.txt

6.894 OODL, Spring’01, Assignment 1, Due Thurs. 2/8/01
Submit via email to oodl-course@ai.mit.edu
Name:
Email:
Previous programming language experience
  (indicate whether user or implementor):
  * Coursework:
   ** projects:
  * Other (industry experience, general hacking):
Have you had exposure to object-oriented programming?
Fluent in the following programming languages:
Interests (why are you here?):

Topics you would most like covered, focussed on: